Author's response to reviews

Title: Genotyping of Streptococcus agalactiae (group B streptococci) isolated from vaginal and rectal swabs of women at 35-37 weeks of pregnancy

Authors:

  Nabil Abdullah El Aila (nabil.elaila@ugent.be)
  Inge Tency (Inge.Tency@UGent.be)
  Geert Claeys (Geert.Claeys@UGent.be)
  Bart Saerens (BartSaerens@gmail.com)
  Rita Verhelst (Rita.Verhelst@UGent.be)
  Ellen De Backer (ellekendb@yahoo.com)
  Marleen Temmerman (Marleen.Temmerman@UGent.be)
  Mario Vaneechoutte (Mario.Vaneechoutte@UGent.be)

Version: 4 Date: 8 September 2009

Author's response to reviews: see over
Sir,

The formatting changes requested have been done accordingly.

Sincerely

Mario Vaneechoutte
Lab. Bacteriol. Research
Univ. Gent
Belgium

Formatting changes requested
----------------------------
Major revisions (We require the author to make these changes)

Footnotes –
We are unable to accommodate footnotes in the full text version of the manuscript. We suggest that the footnotes are incorporated into the text of the manuscript. Alternatively, you may wish to make an appendix section containing your footnotes.
A: We have removed the footer. We think this is what is meant with footnotes.

Author list –
The list of authors in the manuscript should be written exactly as they are in the submission system. The authors (forename, initial (if any), surname) should be written identically and in the order in which they are listed in the submission system.
A: This has been corrected

Affiliations –
Each affiliation should be written in full, in the following format: department, institute, city, and country. This may mean repeating institutional addresses if the authors are from different departments within an institution.
A: We have corrected.

Reference 45 please provide full and accurate details for this reference. This means for an Article providing year/volume/issue/page numbers and for a book the publisher/city/year.
A: This reference is not an article or book, it is guidelines from the Belgian Health Council. I have indicated its website.

Reference 46: please provide full and accurate details for this reference. This means for an Article providing year/volume/issue/page numbers and for a book the publisher/city/year.
A: This reference is not an article or book, it is an annual report, Study Centre for Perinatal Epidemiology (SPE) Flanders. I have indicated its website. It is in Nederlands Language.


Figure order - The first figure referenced in the manuscript text should be Figure 1. The second, figure 2, etc. Please reorder the figure citation so that they are cited in consecutive order.
A: This error has been corrected.

Minor revisions (We can make these changes on resubmission, although it will speed up publication of your manuscript if you do them while making the major changes above).

'El Aila et al.'
Please remove this text from the head of every page
A: It has been removed

Line numbers - Please remove these.
A: It has been removed

Tables - Please do not include vertical lines in your tables.
A: I have removed all lines from tables

Figure cropping - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures
A: We do not manage to correct this. We also do not understand the problem, because there seems to be no white space around the figures. Please correct yourself if necessary. Thanks.

Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions as this is the final proofing stage for your manuscript, and you will not be able to make changes after acceptance.
A: We have checked the final manuscript with ErrNet.